\documentclass{amsart}

\setlength{\parindent}{0pt}
\setlength{\parskip}{10pt}

\author{Course Information, Spring 2004}

\begin{document}

\thispagestyle{empty}

\textbf{MATH 425: Complex Variables}

\textsc{Course Information, Spring 2004}

\textsc{Instructor:} Dennis Snow, snow.1@nd.edu

\textsc{Office Hours:} HAYE 142, T 9:30--11:00, H 11:30--1:00 and by
appointment

\textsc{Text:} Brown \& Churchill, \emph{Complex Variables and
Applications}, 7th Edition, Chapters 1--8. Sections 1--80,83--88

\textsc{Class Time:} MWF 3:00--3:50 in HAYE 229.

\textsc{Exams and Quizzes:}

\begin{quote}\begin{tabular}{ll}
Quiz 1  & Jan~30\\
Quiz 2  & Feb~13\\
Midterm & Feb~27\\
Quiz 3  & Mar~19\\
Quiz 4  & Apr~2\\
Quiz 5  & Apr~23\\
Final   & May~5, 4:15--6:15 P.M.\\
\end{tabular}\end{quote}
The quizzes and midterm exam will be handed out on a Friday, to be
returned on the following Monday. You may take as much time as you wish
and may use your textbook. The room for the final exam will be announced.

\textsc{Honor Code:} The Honor Code is in effect for all exams and
quizzes. Copying in any form or submitting work done by others as your
own is a violation of the Honor Code.

\textsc{Grades:} Grades will be based on a total of 250 points broken
down as follows:

\begin{quote}\begin{tabular}{lcr}
Quizzes &=& 50\\
Midterm     &=& 100\\
Final       &=& 100\\
\end{tabular}\end{quote}

\textsc{Web Page:} Course information, including handouts, messages, and
suggested exercises, can be found at
\begin{quote}
\verb|http://www.nd.edu/~snow/math425.html|.
\end{quote}

\end{document}
